Author's response to reviews

Title: Analysis of EGFR, HER2, and TOP2A gene status and chromosomal polysomy in gastric adenocarcinoma from Chinese patients

Authors:

Zhiyong Liang (liangzhiyong1220@yahoo.com)
Xuan Zeng (zengx88@yahoo.com.cn)
Jie Gao (zy_795@hotmail.com)
Shafei Wu (Suphie@sina.com)
Peng Wang (chinaroc@tom.com)
Xiaohua Shi (shixiaohua3762@hotmail.com)
Jing Zhang (jingzhangsmile@163.com)
Tonghua Liu (liuth_pumch@yahoo.com.cn)

Version: 6 Date: 2 December 2008

Author's response to reviews: see over
Dear Editor,

Thank you very much for email about our manuscript, we have read your suggestions carefully and checked the manuscript again, some changes were made.

**Questions:**

**Major Revisions**

References - References should be cited in the text using consecutive numbers in square brackets. Please do not use superscript numbers. Also, please place the reference before the full stop. For example "text [1]." Tables - Tables must be created using a Tables tool, and should not be linked files, figures, or generated with tabbed text. Tables must be divided into rows and columns and be oriented in a portrait page layout. Please ensure that text alignment in columns is clear and consistent. We recommend the MS Word auto-format template: simple 1. Please do not include vertical lines in your tables.

**Minor Revisions**

Qualifications - Please remove the authors' qualifications (e.g. PhD / MD) from the manuscript and submissions system.

**Answers:**

We have changed references cited in the text by using consecutive numbers in square brackets and placed the references before the full stop. Tables were created using MS Word auto-format template: simple 1.

All the authors' qualifications were removed from the manuscript and submissions system.

Best regards,

Tonghua Liu